\documentclass[onecolumn,letterpaper,11pt]{article}

%===================================================================
\usepackage{bibentry} 
\usepackage[usenames]{color}

\input{../common_latex_definitions}
\input{../common_latex_memo_format}

\nobibliography*             % tell latex no references at the end
%===================================================================

\begin{document}
\wikidoc{file=Memo.RefsCOLREGS}
\wikidoc{refs=false}         % tell texwiki no references at the end
\wikidoc{top_pdf_link=false} % tell texwiki no references at the end

%===================================================================
\section*{References in Related to COLREGS}
%===================================================================

\subsection*{The COLREGS rules themselves}

\begin{packed_itemize}

\item \fullcite{colregs}

\end{packed_itemize}

\subsection*{Very good texts about the COLREGS}

\begin{packed_itemize}

\item \fullcite{allen2004}
\item \fullcite{cockroft2011}

\end{packed_itemize}

\subsection*{Selected COLREGS References from Our Lab}

\begin{packed_itemize}

\item \fullcite{ben17}
\item \fullcite{woerner18autrobots}
\item \fullcite{ben18oceans}
\item \fullcite{woerner18oceans}
\item \fullcite{ben17oceans}
\item \fullcite{woerner17legibility}
\item \fullcite{leavitt2017ms}
\item \fullcite{woerner2016phd}

\end{packed_itemize}

\subsection*{Selected References Related to COLREGS on Autonomous Marine Vehicles}

\begin{packed_itemize}

\item \fullcite{minne2017ms}
\item \fullcite{johansen2016}
\item \fullcite{henderson06}
\item \fullcite{kwzh14}
\item \fullcite{fiorini1993}
\item \fullcite{kuwata2014}
\item \fullcite{kuwata2011}
\item \fullcite{shah2014}
\item \fullcite{snape2011a}
\item \fullcite{vandenberg2008a}

\end{packed_itemize}

\wikinoref

\newpage
\nobibliography{biblio.bib}
\bibliographystyle{plain}

\end{document} 



